POLITECNICO DI TORINO Repository ISTITUZIONALE

Structural Balance via Gradient Flows over Signed Graphs

Original

Structural Balance via Gradient Flows over Signed Graphs / Cisneros-Velarde, P.; Friedkin, N. E.; Proskurnikov, A. V.; Bullo, F.. - In: IEEE TRANSACTIONS ON AUTOMATIC CONTROL. - ISSN 0018-9286. - 66:7(2021), pp. 3169-3183. [10.1109/TAC.2020.3018435]

Availability: This version is available at: 11583/2909894 since: 2021-06-29T11:41:27Z

Publisher: Institute of Electrical and Electronics Engineers Inc.

Published DOI:10.1109/TAC.2020.3018435

Terms of use:

This article is made available under terms and conditions as specified in the corresponding bibliographic description in the repository

Publisher copyright IEEE postprint/Author's Accepted Manuscript

©2021 IEEE. Personal use of this material is permitted. Permission from IEEE must be obtained for all other uses, in any current or future media, including reprinting/republishing this material for advertising or promotional purposes, creating new collecting works, for resale or lists, or reuse of any copyrighted component of this work in other works.

(Article begins on next page)

Structural Balance via Gradient Flows over Signed Graphs

Pedro Cisneros-Velarde, Student Member, IEEE, Noah E. Friedkin, Anton V. Proskurnikov, Senior Member, IEEE, Francesco Bullo, Fellow, IEEE

Abstract—Structural balance is a classic property of signed graphs satisfying Heider's seminal axioms. Mathematical sociologists have studied balance theory since its inception in the 1940s. Recent research has focused on the development of dynamic models explaining the emergence of structural balance. In this paper, we introduce a novel class of parsimonious dynamic models for structural balance based on an interpersonal influence process. Our proposed models are gradient flows of an energy function, called the dissonance function, which captures the cognitive dissonance arising from the violations of Heider's axioms. Thus, we build a new connection with the literature on energy landscape minimization. This gradient-flow characterization allows us to study the transient and asymptotic behaviors of our model. We provide mathematical and numerical results describing the critical points of the dissonance function.

I. INTRODUCTION

1) Problem description and motivation: Signed graphs represent networked systems with interactions classified as positive or negative, e.g., cooperation or antagonism, promotion or inhibition, attraction or repulsion. Such graphs naturally arise in diverse fields, e.g., political science [14], communication studies [19] and biology [20]. In sociology [6], [9], they are used to represent friendly or antagonistic relationships, whereby signed edges may be interpreted as interpersonal sentiment appraisals. In the work by Heider [12], each individual appraises all other individuals either positively (friends, allies) or negatively (enemies, rivals). Heider postulated four famous axioms: (i) "the friend of a friend is a friend," (ii) "the enemy of a friend is an enemy," (iii) "the friend of an enemy is an enemy," and (iv) "the enemy of an enemy is a friend." Violations of these axioms lead to cognitive tensions and dissonances that the individuals strive to resolve; in this sense, Heider's axioms are consistent with the general theory of cognitive dissonance [8]. A signed network satisfying Heider's axioms is called structurally balanced and can have only two possible configurations: either all of its members have positive relationships with each other and become a unique faction, or

Pedro Cisneros-Velarde (pacisne@gmail.com), Noah E. Friedkin and Francesco Bullo ({friedkin,bullo}@ucsb.edu) are with the Center for Control, Dynamical Systems and Computation, University of California, Santa Barbara.

Anton V. Proskurnikov is with the Department of Electronics and Telecommunications, Politecnico di Torino, Turin, Italy and also with the Institute for Problems of Mechanical Engineering, Russian Academy of Sciences, St. Petersburg, Russia. there exist two factions in which members of the same faction are friends but enemies with every other member in the other faction. We refer to [6], [9] for textbook treatment and to [29] for a recent comprehensive survey.

Whereas Heider's theory describes the qualitative emergence of structural balance as the result of tension-resolving cognitive mechanisms, it does not provide a quantitative description of these mechanisms and dynamic models explaining the emergence of balance. The aim to fill this gap has given rise to the important research area of dynamic structural balance. The Kułakowski et al. [17] model postulates an influence process, whereby any individual *i* updates her appraisal of individual *j* based on what others positively or negatively think about *j*. The Traag et al. [27] model postulates a homophily process, whereby any individual i updates her appraisal of jaccording to how much she agrees with j on the appraisals of their common acquaintances. Both models explain convergence to structural balance under certain assumptions on the initial state (see below for more information). Remarkably, both models assume the existence of so-called *self-appraisals* (loops in the signed graph) that strongly influence the system dynamics. Self-appraisals can be interpreted as individuals' positive or negative opinions of themselves.

A second line of research, consistent with dissonance theory, has focused on formulating social balance via appropriate energy functions. The work [23] proposes an energy function for binary appraisal matrices with global minima that represent structurally stable configurations; it is argued that a dynamic structural balance model should aim to navigate through this energy landscape and look for its minima. Some models (e.g., [2], [3]) were designed precisely to achieve this task. The work [7] computes a distance to balance via a combinatorial optimization problem, inspired by Ising models.

The purpose of this paper is threefold. First, we aim to propose a more parsimonious model of the influence process establishing structural balance, that is, a model without selfappraisal weights. Our argument for dropping these variables is that balance theory axioms do not include self-appraisals, and the inclusion of such appraisals amounts to an additional assumption and introduces unnecessary complexities. Second, we aim to connect the literature on dynamic structural balance with the literature treating social balance as an optimization problem. Finally, we aim to emphasize through numerical simulations that our parsimonious model does not suffer from a key limitation present in the Kułakowski et al. model, namely that the Kułakowski et al. model cannot predict the emergence of structural balance from asymmetric initial configurations.

This work is supported by the U. S. Army Research Laboratory and the U. S. Army Research Office under grant number W911NF-15-1-0577. The views and conclusions contained in this document are those of the authors and should not be interpreted as representing the official policies, either expressed or implied, of the Army Research Laboratory or the U.S. Government. A. Proskurnikov acknowledges financial support of RFBR grant 20-01-00619.

2) Further comments on the state of the art: We now present a summary of the current literature on dynamic structural balance. Historically, the first models appeared in the physics community [2], [3], [25]. These models borrowed some concepts from statistical physics and had the particularity of assuming that the appraisals between individuals are binary valued (either +1 or -1). At the same time, they rely on hardwired random mechanisms for the asynchronous updates of the appraisals that lack a sociological insightful interpretation.

Another type of proposed models is based on discreteand continuous-time dynamical systems with real-valued appraisals. The seminal models of this kind are due to Kułakowski et al. [17] (later analyzed more formally by [22]) and Traag et al. [27]. Models with real-valued appraisals capture not only signs, but also magnitudes of positive or negative sentiments. All these models adopt synchronous updating and stipulate sociological meaningful rules for the updating of appraisals, based on either influence or homophily processes. The following facts are known about the Kułakowski et al. influence-based and the Traag et al. homophily-based models: the set of well-behaved initial conditions that lead the social network towards social balance for the first model is a subset of the set of normal matrices, while the second model can work under generic initial conditions. Similar results are obtained by [24] for two discrete-time models based on influence and homophily respectively: influence-based processes do not perform well under generic initial conditions (in contrast to the homophily-based processes). Finally, only the models proposed in [24] and a variation of the model by Kułakowski et al. proposed in the early work [17], have a bounded evolution of appraisals, whereas the others have finite escape time.

Recent work has also started to focus on dynamic models for other relevant configuration of signed graphs, e.g., configurations that satisfy only a subset of the four Heider's axioms. The work [10] provides a parsimonious model explaining the emergence of a generalized version of structural balance from any initial configuration; this model is based on an influence process of positive contagion whereby influence is accorded only to positively-appraised individuals. A second model in this area is proposed by [16]. Finally, there has been a third type of models that propose the emergence of structural balance or other generalized balance structures for undirected graphs from a game theoretical perspective [5], [21], [28].

3) Contributions: First of all, we contribute by proposing two new dynamic models that do not adopt the long-standing assumption of self-appraisals and describe the evolution of signed networks without self-loops. We argue that the introduction of self-weights is poorly justified and that a model without them is a more faithful representation of Heider's theory. The first model, called the *pure-influence model*, is a modification of the classic model by Kułakowski et al. which is obtained by eliminating self-appraisals (and thus reducing the system's dimension). Analysis of its convergence properties reduces to the analysis of our second model, called the *projected pure-influence model*, which arises as a projection of the first model onto the unit sphere. This second model has a self-standing interest, since it enjoys bounded evolution of the appraisals, while the first model shares the finite escape time property of the classic model by Kułakowski et al.

Our second contribution is to build a bridge between dynamic structural balance and structural balance as an optimization problem. We propose an energy function inspired by [23], namely the dissonance function, which measures the degree at which Heider's axioms are violated among the individuals of a social network. We show that this energy function has global minima that correspond to signed graphs satisfying structural balance in the case of real-valued appraisals (restricted on the unit sphere). Moreover, we show that our (projected) pureinfluence model is the gradient system of the dissonance function in the case of undirected signed graphs, and hence the critical points of the dissonance function are the equilibria of our dynamical system. Thus, we establish a novel connection between dynamic structural balance and the characterization of structural balance as the minima of an energy function. Remarkably, our derivations show that this property of our models is enabled by the elimination of self-appraisals. Thus, the models contributed in this paper may be considered as both an interpersonal influence process and an extremum seeking dynamics for the dissonance function.

Our third and more detailed contribution is the mathematical analysis of the projected pure-influence model in the cases where the initial appraisal matrix is symmetric. In particular, we provide a complete characterization of the critical points of the dissonance function (i.e., the equilibrium points of the projected pure-influence model). This characterization relies upon a special submanifold of the Stiefel manifold and its properties. Along with the characterization of the critical points, we analyze their local stability properties and provide some results on convergence towards structural balance.

Our final contribution is a Monte Carlo numerical study of the convergence of our models to structural balance under generic initial conditions in both the symmetric and the asymmetric case. For the symmetric case, our numerical result is comparable to, but stronger than, what has already been proved for the Kułakowski et al. model: our models converge to structural balance under generic symmetric initial conditions. One key advantage of our models, as compared with those by Kułakowski et al., is that convergence to structural balance emerges under generic asymmetric initial conditions. Based on these numerical results, we formulate relevant conjectures.

4) Paper organization: Section II presents preliminary concepts. Section III presents our models and shows they are gradient flows. Section IV and Section V contain an analysis of equilibria and important convergence results, respectively. Section VI contains numerical results and conjectures. Finally, Section VII contains some concluding remarks.

II. PRELIMINARIES

A. Signed weighted digraphs

Given an $n \times n$ matrix $X = (x_{ij})$ with entries taking values in $[-\infty, \infty]$, let G(X) denote the signed directed graph where the directed edge $i \to j$ exists if and only if $x_{ij} \neq 0$, and x_{ij} represents its signed weight. The directed graph G(X)is complete if X has no zero entries, except for the main diagonal. G(X) has no self-loops if and only if X has zero diagonal entries. Let x_{i*} denote the *i*th row of the matrix X and x_{*i} the *i*th column of the matrix X. Let $sign(X) = (sign(x_{ij}))$, where $sign : [-\infty, \infty] \rightarrow \{-1, 0, +1\}$ is as usual

$$\operatorname{sign}(x) = \begin{cases} -1, & \text{if } x < 0, \\ 0, & \text{if } x = 0, \\ +1, & \text{if } x > 0. \end{cases}$$

Given a sequence a_1, \ldots, a_n , let $B = \text{diag}(a_1, \ldots, a_n)$ denote the diagonal $n \times n$ matrix (b_{ij}) , where $b_{ii} = a_i$ and $b_{ij} = 0$ for $i \neq j$. For an $n \times n$ matrix X, define $\text{diag}(X) = \text{diag}(x_{11}, \ldots, x_{nn})$. For a vector $v \in \mathbb{R}^n$, define $\text{diag}(v) = \text{diag}(v_1, \ldots, v_n)$. Let \mathbb{O}_n denote the $n \times 1$ vector of zeros, and $\mathbb{O}_{n \times n}$ the $n \times n$ matrix with zero entries.

Let \succ and \prec denote "entry-wise greater than" and "entrywise less than," respectively.

A triad (if it exists) is a cycle between three nodes in G(X). The sign of a triad is defined by the sign of the product of the weights composing a triad. For example, the triad $i \to j \to k \to i$ has sign sign $(x_{ij}x_{jk}x_{ki})$.

A real-valued matrix Z is *irreducible* if its graph G(Z) is strongly connected (a directed path between every two nodes exists) and *reducible* otherwise. If Z is reducible, a permutation matrix P exists such that the matrix

$$PZP^{\top} = \begin{bmatrix} Z_1 & * & \dots & * \\ 0 & Z_2 & \dots & * \\ \vdots & & & \\ 0 & & & Z_k \end{bmatrix}$$

is upper-triangular with irreducible blocks Z_i (some of them can be 1×1 matrices). If $Z = Z^{\top}$, the latter matrix is blockdiagonal matrix $PZP^{\top} = \text{diag}(Z_1, \ldots, Z_k)$ and the graphs $G(Z_i)$ are the *connected components* of the graph G(Z).

B. Sets of matrices and the Frobenius inner product

Given two matrices $A, B \in \mathbb{R}^{n \times n}$, their Frobenius inner product is defined by $\langle\!\langle A, B \rangle\!\rangle_F = \operatorname{trace}(B^\top A)$; the inducednorm is $||A||_F = \sqrt{\langle\!\langle A, A \rangle\!\rangle_F}$. Some important properties for the trace operator are: $\operatorname{trace}(A) = \operatorname{trace}(A^\top)$, $\operatorname{trace}(AB) =$ $\operatorname{trace}(BA)$, and, for all $d \in \mathbb{N}$, $\operatorname{trace}(A^d) = \sum_{i=1}^n \lambda_i^d$ where $\lambda_1, \ldots, \lambda_n$ are the eigenvalues of A.

Let $\mathbb{R}_{\text{zero-diag}}^{n \times n}$ be the set of $n \times n$ real matrices with zero diagonal entries, and $\mathbb{R}_{\text{zero-diag},\text{symm}}^{n \times n}$ be the set of symmetric matrices belonging to $\mathbb{R}_{\text{zero-diag}}^{n \times n}$. Let $\mathbb{S}^{n \times n}$ be the unit sphere in $\mathbb{R}^{n \times n}$, that is $A \in \mathbb{S}^{n \times n}$ if and only if $A \in \mathbb{R}^{n \times n}$ with $\|A\|_F = 1$. Similarly, we define the sets $\mathbb{S}_{\text{zero-diag}}^{n \times n} = \mathbb{R}_{\text{zero-diag},\text{symm}}^{n \times n} \cap \mathbb{S}^{n \times n}$. Let $\mathbb{R}_{\text{zero-diag},\text{symm}}^{n \times n} \cap \mathbb{S}^{n \times n}$. Let $\mathbb{R}_{\text{diag}}^{n \times n}$ be the set of all real diagonal matrices and $\mathbb{S}_{\text{zero-diag}}^{n \times n}$.

Let $\mathbb{R}^{n \times n}_{\text{diag}}$ be the set of all real diagonal matrices and $\mathbb{R}^{n \times n}_{\text{sk-symm}}$ be the set of all skew-symmetric matrices. Then, we have the following orthogonal decomposition of $\mathbb{R}^{n \times n}$ equipped with the Frobenius inner product:

$$\mathbb{R}^{n \times n} = \mathbb{R}^{n \times n}_{\text{sk-symm}} \oplus \mathbb{R}^{n \times n}_{\text{zero-diag,symm}} \oplus \mathbb{R}^{n \times n}_{\text{diag}}.$$
 (1)

C. A review on structural balance

Throughout the paper we deal with social networks composed of $n \geq 3$ individuals, although the definition of structural balance (Definition II.3) is formally applicable to the case of degenerate networks with n = 1 or n = 2 nodes.

Definition II.1 (Appraisal matrix and network). We let the entry x_{ij} of the matrix $X \in \mathbb{R}^{n \times n}$ denote the appraisal (or qualitative evaluation) held by individual *i* of individual *j*. The sign of x_{ij} indicates if the relationship is positive (+1), negative (-1) or of indifference (0). The magnitude of x_{ij} indicates the strength of the relationship. x_{ii} can be interpreted as *i*'s self-appraisal. We call X the appraisal matrix, and G(X) the appraisal network.

Definition II.2 (Heider's axioms and social balance notions). *The* Heider's axioms *are*

H1) A friend of a friend is a friend,

H2) An enemy of a friend is an enemy,

H3) A friend of an enemy is an enemy,

H4) An enemy of an enemy is a friend.

An appraisal network G(X) is structurally balanced in Heider's sense, if it is complete and satisfies axioms H1)-H4).

Consider a complete appraisal network G(X). We call a *faction* any group of agents whose members positively appraise each other. We say two factions are *antagonistic* if every representative from one faction negatively appraise every representative of the other faction. It can be shown [4], [11], [12] that Heider's structural balance condition for G(X)with $n \ge 3$ nodes holds if and only if either the individuals constitute a single faction or can be partitioned into two antagonistic factions. The possession of the latter property may thus be considered as an alternative definition of structural balance (and is formally applicable to graphs without triads).

Definition II.3 (Structural balance). A complete appraisal network G(X) is said to satisfy structural balance, if G(X) is composed by one faction or two antagonistic factions; or, whenever $n \ge 3$, equivalently, that all triads are positive, i.e., $x_{ij}x_{jk}x_{ki} > 0$ for any different $i, j, k \in \{1, ..., n\}$.

Notice that a structurally balanced graph is always signsymmetric: $sign(x_{ij}) = sign(x_{ji})$ for any $i \neq j$. For simplicity we will say that a matrix X corresponds to structural balance whenever G(X) satisfies structural balance.

III. PROPOSED MODELS AND REPRESENTATION AS GRADIENT FLOWS

In this section we propose our models defining them over the set of symmetric matrices. We postponed the general asymmetric setting to Section VI.

A. Pure-influence model

We propose our new dynamic model solely based on interpersonal appraisals.

Definition III.1 (Pure-influence model). *The* pure-influence model *is a system of differential equations on the set of zerodiagonal matrices* $\mathbb{R}^{n \times n}_{\text{zero-diag}}$ *defined by*

$$\dot{x}_{ij} = \sum_{\substack{k=1\\k\neq i,j}}^{n} x_{ik} x_{kj},$$
(2)

for any $i, j \in \{1, ..., n\}$ and $i \neq j$. Here $x_{ij}, i \neq j$, are the off-diagonal entries of a zero-diagonal matrix $X \in \mathbb{R}^{n \times n}_{\text{zero-diag}}$. In equivalent matrix form, the previous equations read:

$$\dot{X} = X^2 - \operatorname{diag}(X^2), \qquad X(0) \in \mathbb{R}^{n \times n}_{\operatorname{zero-diag}}.$$
 (3)

We interpret X as the interpersonal appraisal matrix. While system (2) does not define the evolution of self-appraisals, the matrix reformulation (3) ensures $\operatorname{diag}(\dot{X}) = \mathbb{O}_{n \times n}$ and, since $X(0) \in \mathbb{R}^{n \times n}_{\text{zero-diag}}$ means $\operatorname{diag}(X(0)) = \mathbb{O}_{n \times n}$, we have $\operatorname{diag}(X(t)) = \mathbb{O}_{n \times n}$ for all positive times t.

Our model is a modification of the classical model proposed by Kułakowski et al. [17], where self-appraisals play a crucial role in the dynamics of the interpersonal appraisals.

Definition III.2 (Kułakowski et al. model). *The* Kułakowski et al. model *is a system of differential equations on the state space* $\mathbb{R}^{n \times n}$ *defined by*

$$\dot{x}_{ij} = \sum_{k=1}^{n} x_{ik} x_{kj} = x_{ij} (x_{ii} + x_{jj}) + \sum_{\substack{k=1\\k \neq i,j}}^{n} x_{ik} x_{kj}, \quad (4a)$$

$$\dot{x}_{ii} = x_{ii}^2 + \sum_{\substack{k=1\\k\neq i}}^n x_{ik} x_{ki},$$
(4b)

for any $i \neq j \in \{1, ..., n\}$. In equivalent matrix form, the previous equations read: $\dot{X} = X^2$.

Remark III.1 (The problem with self-appraisals). The introduction of self-appraisals in model (4) is objectionable on several grounds. The first conceptual problem is that selfappraisals are not considered in any definition of structural balance in the social sciences. Heider's axioms in Definition II.2 do not take into account self-appraisals: social balance is a function of only interpersonal appraisals. Moreover, once self-appraisals are introduced, one needs to postulate why and how self-appraisals affect interpersonal appraisals, *i.e.*, justify the choice of the first addendum for the right hand side of (4a). Finally, one needs to postulate how they evolve, i.e., justify the choice for the right hand side of (4b). In summary, the pure influence model (2) avoids these difficulties and stays closer to the foundations of structural balance, in which individuals are attending only to interpersonal appraisals. Even though $\dot{X} = X^2$ may appear mathematically simpler or more elegant than $\dot{X} = X^2 - \text{diag}(X^2)$, we believe the latter model is actually more parsimonious, lower dimensional, and more faithful to Heiders' axioms.

One easily notices the following important property of the pure-influence model (3): the right-hand side is an analytic function of X so that the equation enjoys (local) existence and uniqueness of the solutions. A second property is that, if $X(0) = X(0)^{\top}$, then $X(t) = X(t)^{\top}$ for all subsequent times. This implies that the pure-influence model is well defined over the set of symmetric (zero diagonal) matrices $\mathbb{R}_{\text{zero-diag,symm}}^{n \times n}$.

B. Dissonance function

We introduce and study the properties of a useful *dissonance function* that summarize the total amount of cognitive dissonances [8] among the members of a social network due to the

lack of satisfaction of Heider's axioms. Recall that, according to Definition II.3, a triad $i \rightarrow j \rightarrow k \rightarrow i$ satisfies the axioms if and only if $x_{ij}x_{jk}x_{ki} > 0$.

Definition III.3 (Dissonance function). *The* dissonance function $\mathcal{D}: \mathbb{R}^{n \times n}_{\text{zero-diag}} \to \mathbb{R}$ *is*

$$\mathcal{D}(X) = -\sum_{\substack{i,j,k=1\\i\neq j, j\neq k, k\neq i}}^{n} x_{ij} x_{jk} x_{ki} = -\operatorname{trace}(X^3) = -\sum_{i=1}^{n} \lambda_i^3,$$
(5)

where $\{\lambda_i\}_{i=1}^n$ is the set of eigenvalues of X.

We plot \mathcal{D} in a low-dimensional setting in Figure 1.



Fig. 1. For n = 3, an arbitrary symmetric unit-norm zero-diagonal matrix $X \in \mathbb{S}_{zro-diag,symm}^{n \times n}$ is described by (x_{12}, x_{23}, x_{31}) with these coordinates living in the sphere $x_{12}^2 + x_{23}^2 + x_{31}^2 = 1$. In the upper figure, we plot this sphere with a heatmap, with dark blue being the lowest value and light yellow the largest value, according to the evaluation of the dissonance function $\mathcal{D}(X)$. The function has four global minima corresponding to the four possible configurations of G(X) satisfying structural balance, and we can qualitatively appreciate the convergence of solution trajectories to these minima in the superimposed vector field on the sphere. The lower figure is a stereographic projection of the upper figure.

Energy landscapes in social balance theory are studied in [7], [23]. Our proposed dissonance function is the extension to $\mathbb{R}_{\text{zero-diag}}^{n \times n}$ of the energy function proposed by [23] for the setting of binary-valued symmetric appraisal matrices. For binary-valued appraisals, the global minima of \mathcal{D} correspond to networks that satisfy structural balance, since all triads are positive (Definition II.3). Thus, \mathcal{D} naturally measures to which extent Heider's axioms are violated in a complete graph. **Lemma III.2** (Properties of the dissonance function). Consider the dissonance function \mathcal{D} and pick $X \in \mathbb{R}^{n \times n}_{\text{zero-diag}}$. Then (i) \mathcal{D} is analytic and attains its maximum and minimum

- values on any compact matrix subset of $\mathbb{R}_{\text{zero-diag}}^{n \times n}$,
- (ii) if G(X) satisfies structural balance, then $\mathcal{D}(X) = 0$,

(iii) $\mathcal{D}(X) = \mathcal{D}(X^{\top}),$ (iv) $\mathcal{D}(X) = -\langle \langle X^2, X^{\top} \rangle \rangle_F.$ Additionally, if $||X||_F = 1$, that is, $X \in \mathbb{S}^{n \times n}_{\text{zero-diag}}$, then (v) $-1 \leq \mathcal{D}(X) \leq 1.$

Proof. Here we show only property (v), since the other properties follow easily from the definition of \mathcal{D} . We note:

$$\begin{split} \left\|X^{2}\right\|_{F}^{2} &= \sum_{i,j=1}^{n} (X^{2})_{ij}^{2} = \sum_{i,j=1}^{n} (X_{i*}X_{*j})^{2} \\ &\leq \sum_{i,j=1}^{n} \|X_{i*}\|_{2}^{2} \|X_{*j}\|_{2}^{2} = \left(\sum_{i=1}^{n} \|X_{i*}\|_{2}^{2}\right) \left(\sum_{j=1}^{n} \|X_{*j}\|_{2}^{2}\right) \\ &= \left(\sum_{i,k=1}^{n} x_{ik}^{2}\right)^{2} = \|X\|_{F}^{2} = 1. \end{split}$$

Now, note that the Frobenius norm on the set of matrices coincides with the Euclidean norm of a single vector obtained by stacking the column vectors of the matrix. Then, by the Cauchy-Schwarz inequality applied to the innerproduct $\langle\!\langle \cdot, \cdot \rangle\!\rangle_F$, it follows that: $|D(X)| = |\langle\!\langle X^2, X \rangle\!\rangle_F| \leq ||X^2||_F ||X||_F \leq (||X||_F)^3 \leq 1$ when $||X||_F \leq 1$.

C. Transcription on the unit sphere and the projected pureinfluence model

We start by noting a simple fact. Given a trajectory $X : \mathbb{R}_{\geq 0} \to \mathbb{R}_{\text{zero-diag}}^{n \times n} \setminus \{\mathbb{O}_{n \times n}\}$, there exist unique trajectories $\eta : \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$ and $Z : \mathbb{R}_{\geq 0} \to \mathbb{S}_{\text{zero-diag}}^{n \times n}$ such that $X(t) = \eta(t)\mathcal{Z}(t)$, where $\eta(t) = ||X(t)||_F$ and $\mathcal{Z}(t) = X(t)/||X(t)||_F$.

Theorem III.3 (Transcription of the pure-influence model). The pure-influence model (2) with initial conditions in $\mathbb{R}^{n \times n}_{\text{zero-diag,symm}}$ can be expressed as the following system of differential equations:

$$\begin{aligned} \dot{\mathcal{Z}} &= \eta \mathcal{P}_{\mathcal{Z}^{\perp}}(\mathcal{Z}^2 - \operatorname{diag}(\mathcal{Z}^2)) \\ &= \eta(\mathcal{Z}^2 - \operatorname{diag}(\mathcal{Z}^2) + \mathcal{D}(\mathcal{Z})\mathcal{Z}), \end{aligned}$$
(6a)

$$\dot{\eta} = -\mathcal{D}(\mathcal{Z})\eta^2,\tag{6b}$$

where $\eta : \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$ and $\mathcal{Z} : \mathbb{R}_{\geq 0} \to \mathbb{S}^{n \times n}_{\text{zero-diag,symm}}$. Here $\mathcal{P}_{\mathcal{Z}^{\perp}}$ is the orthogonal projection onto $\operatorname{span}\{\mathcal{Z}\}^{\perp}$ in the vector space of square matrices with the Frobenius inner product.

Proof. Since $\dot{X} = \dot{\eta}Z + \eta \dot{Z}$ and $X^2 - \text{diag}(X^2) = \eta^2 (Z^2 - \text{diag}(Z^2))$, equation (3) can be written as

$$\dot{\eta}\mathcal{Z} + \eta\dot{\mathcal{Z}} = \eta^2 \left(\mathcal{Z}^2 - \operatorname{diag}(\mathcal{Z}^2)\right).$$
 (7)

Differentiating the equality $\|Z(t)\|_F^2 = \langle\!\langle Z(t), Z(t) \rangle\!\rangle_F = 1$, one shows that $\langle\!\langle Z(t), \dot{Z}(t) \rangle\!\rangle_F = 0$, that is, $Z(t) \perp \dot{Z}(t)$. Computing the Frobenius inner product with Z(t) on both sides of (7), equation (6b) is immediate:

$$\dot{\eta} = \eta^2 \langle\!\langle \mathcal{Z}(t), \mathcal{Z}^2(t) - \operatorname{diag}(\mathcal{Z}^2(t)) \rangle\!\rangle_F = \eta^2 \langle\!\langle \mathcal{Z}(t), \mathcal{Z}^2(t) \rangle\!\rangle_F = -\mathcal{D}(\mathcal{Z}(t)) \eta^2.$$
(8)

where we have used the fact that $\mathcal{Z}(t)$ is symmetric, and that $\operatorname{diag}(\mathcal{Z}(t)) = \mathbb{O}_{n \times n}$ and hence $\langle\!\langle \mathcal{Z}(t), \operatorname{diag}(\mathcal{Z}^2(t)) \rangle\!\rangle_F =$ $\operatorname{trace}(\mathcal{Z}(t)^\top \operatorname{diag}(\mathcal{Z}^2(t))) = 0$. Substituting (8) into equation (7), one arrives at $\dot{\mathcal{Z}} = \eta \left(\mathcal{Z}^2 - \operatorname{diag}(\mathcal{Z}^2) + \mathcal{D}(\mathcal{Z}) \right)$.

Given $Y \in \mathbb{R}^{n \times n}$, let $\mathcal{P}_{\mathcal{Z}}(Y) = \langle\!\langle Y, \mathcal{Z} \rangle\!\rangle_F \mathcal{Z}$, i.e., $\mathcal{P}_{\mathcal{Z}}$ is the orthogonal projection operator onto the linear space spanned by \mathcal{Z} ; and let $\mathcal{P}_{\mathcal{Z}^{\perp}}(Y) = Y - \mathcal{P}_{\mathcal{Z}}(Y) = Y - \langle\!\langle Y, \mathcal{Z} \rangle\!\rangle_F \mathcal{Z}$ be the orthogonal projection onto the space perpendicular to the linear space spanned by \mathcal{Z} . Then, we observe that $\mathcal{P}_{\mathcal{Z}^{\perp}}(\mathcal{Z}) = 0$ and $\mathcal{P}_{\mathcal{Z}^{\perp}}(\mathcal{Z}) = \dot{\mathcal{Z}}$. Using these results, we apply $\mathcal{P}_{\mathcal{Z}^{\perp}}$ to both sides of (7) and obtain $\dot{\mathcal{Z}} = \eta \mathcal{P}_{\mathcal{Z}^{\perp}}(\mathcal{Z}^2 - \operatorname{diag}(\mathcal{Z}^2))$. This concludes the proof of equations (6).

In what follows, we are primarily interested in the dynamics (6a), describing the behavior of the bounded component $\mathcal{Z}(t)$. From Lemma A.1 we observe that η is a time-scale change for (6a) and so, for our convenience, we get rid of it and obtain the following dynamical system on the unit sphere.

Definition III.4 (Projected pure-influence model). *The* projected pure-influence model *is a system of differential equations on the manifold* $\mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$ *defined by*

$$\dot{Z} = Z^2 - \operatorname{diag}(Z^2) + \mathcal{D}(Z)Z.$$
(9)

Given a solution Z(t) to (9) with initial condition Z(0), Lemma A.1 in the Appendix shows that Z(t) is a time-scaled version of a solution $\mathcal{Z}(t)$ to (6a) with initial condition $\mathcal{Z}(0) = Z(0)$, where η in (6b) can have any positive initial condition. Therefore, there is a solution X(t) to (3) that is both a scaled and time-scaled version of Z(t).

Similarly, projecting onto the unit sphere leads to a new model based on the Kułakowski et al. model.

Definition III.5 (Projected Kułakowski et al. model). *The* projected Kułakowski et al. model *is a system of differential* equations on the manifold $\mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$ defined by

$$\dot{Z}(t) = Z^2 + \mathcal{D}(Z)Z. \tag{10}$$

D. Pure-influence is the gradient flow of the dissonance function

We now let grad \mathcal{D} denote the gradient vector field of the dissonance function \mathcal{D} on the manifold $\mathbb{R}^{n \times n}_{\text{zero-diag}}$ equipped with the Riemannian metric tensor $\langle\!\langle \cdot, \cdot \rangle\!\rangle_F$. We also let $\mathcal{D}|_{\mathbb{S}^{n \times n}_{\text{zero-diag,symm}}}$ denote the restriction of \mathcal{D} onto the manifold $\mathbb{S}^{n \times n}_{\text{zero-diag,symm}}$. We now present the first of our main results.

Theorem III.4 (The pure-influence models over symmetric matrices are gradient flows). Consider the pure-influence model (2) with $X(0) \in \mathbb{R}^{n \times n}_{\text{zero-diag,symm}}$ and the projected pure-influence model (9) with $Z(0) \in \mathbb{S}^{n \times n}_{\text{zero-diag,symm}}$. Then (i) $t \mapsto X(t)$ remains in the set $\mathbb{R}^{n \times n}_{\text{zero-diag,symm}}$ and

$$\dot{X} = -\frac{1}{3} \operatorname{grad} \mathcal{D}(X), \tag{11}$$

(ii) $t \mapsto Z(t)$ remains in the set $\mathbb{S}^{n \times n}_{\text{zero-diag,symm}}$ and

$$\dot{Z} = -\frac{1}{3} \mathcal{P}_{Z^{\perp}} \big(\operatorname{grad} \mathcal{D}(Z) \big) = -\frac{1}{3} \operatorname{grad} \mathcal{D} \Big|_{\mathbb{S}^{n \times n}_{\operatorname{zero-diag,symm}}} (Z).$$
(12)

In other words, the projected pure-influence model (9) is, modulo a constant factor, the gradient flow of the dissonance function \mathcal{D} restricted to the manifold of zero-diagonal unitnorm symmetric matrices $\mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$.

Proof of Theorem 111.4. The forward invariance of the set of symmetric matrices in both statements is immediate from the solution uniqueness. To prove equation (12), we adopt the slight abuse of notation $\operatorname{grad} \mathcal{D}(Z) = \operatorname{grad} \mathcal{D}\Big|_{\mathbb{S}^{n \times n}_{\operatorname{zero-diag,symm}}}(Z)$. With this notation, $Z \mapsto \operatorname{grad} \mathcal{D}(Z)$ is [13, pages 15-17] the unique vector field on $\mathbb{S}^{n \times n}_{\operatorname{zero-diag,symm}}$ such that

$$\frac{d}{dt}\mathcal{D}(Z(t)) = \langle\!\langle \operatorname{grad} \mathcal{D}(Z(t)), \dot{Z}(t) \rangle\!\rangle_F$$
(13)

for any differentiable $Z : [0, \infty) \to \mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$. Here, both $\operatorname{grad} \mathcal{D}(Z(t))$ and $\dot{Z}(t)$ belong to the tangent space to the manifold $\mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$. Now, using the various properties of the trace inner product (e.g., $\dot{Z}(t) \perp Z(t)$), we compute

$$\begin{split} \dot{\mathcal{D}}(Z(t)) &= -(\operatorname{trace}(\dot{Z}(t)Z(t)Z(t)) + \operatorname{trace}(Z(t)\dot{Z}(t)Z(t))) \\ &+ \operatorname{trace}(Z(t)Z(t)\dot{Z}(t)) \\ &= -3\operatorname{trace}(\dot{Z}(t)Z^{2}(t)) = -3\langle\!\langle\dot{Z}(t), Z^{2}(t)\rangle\!\rangle_{F} \\ &= -3\langle\!\langle\dot{Z}(t), Z^{2}(t) - \operatorname{diag}(Z^{2}(t)) + \mathcal{D}(Z(t))Z(t)\rangle\!\rangle_{F}. \end{split}$$

Recalling that $Z^2 - \operatorname{diag}(Z^2) + \mathcal{D}(Z)Z \stackrel{\text{(6a)}}{=} P_{Z^{\perp}}(Z^2 - \operatorname{diag}(Z^2))$ belongs to the tangent space to the manifold $\mathbb{S}_{\operatorname{zero-diag,symm}}^{n \times n}$ at the point Z(t), one arrives at the equality

$$\operatorname{grad} \mathcal{D}(Z) = -3(Z^2 - \operatorname{diag}(Z^2) + \mathcal{D}(Z)Z).$$

This concludes the proof of statement (ii). Finally, equation (11) can be proved in a similar way. \Box

IV. CLASSIFICATION OF SYMMETRIC EQUILIBRIA

We here give the complete classification of the symmetric equilibria in the projected pure-influence model (9); the classification of general asymmetric equilibria remains an open problem. Thanks to Theorem III.4, all symmetric equilibria of the projected pure-influence model are critical points of the dissonance function \mathcal{D} . We start with the equilibrium equation:

$$Z^{2} + \mathcal{D}(Z)Z - \operatorname{diag}(Z^{2}) = \mathbb{O}_{n \times n}, \quad Z \in \mathbb{S}^{n \times n}_{\operatorname{zero-diag,symm}}.$$
(14)

Note that the equilibria Z^* with $\mathcal{D}(Z^*) = 0$ correspond to equilibria of the original system (3) $X(t) \equiv X^* = \eta(0)Z^*$, whereas the others with $\mathcal{D}(Z^*) \neq 0$ lead to

$$X(t) = \eta(t)Z^*, \quad \eta(t) = \frac{\eta(0)}{1 + t\eta(0)\mathcal{D}(Z^*)}$$

defined for $t \in [0, \frac{1}{\eta(0)\mathcal{D}(Z^*)})$ if $\mathcal{D}(Z^*) < 0$ (for which the solution is unbounded) or for $t \ge 0$ if $\mathcal{D}(Z^*) > 0$.

A. Normalized Stiefel matrices

To start with, we introduce a special important manifold of non-square matrices that we will use throughout the paper.

Definition IV.1 (Normalized Stiefel matrices). A matrix $V \in \mathbb{R}^{n \times k}$, for $k \leq n$, is normalized Stiefel (nSt), if

- (i) the columns of V are pairwise orthogonal unit vectors,
 i.e., V[⊤]V = I_k;
- (ii) the norm of each row is the same (obviously, it must be $\sqrt{k/n} \leq 1$): diag $(VV^{\top}) = n^{-1}kI_n$.

Let $nSt(n,k) \subseteq \mathbb{R}^{n \times k}$ denote the set of normalized Stiefel matrices.

In general, the rows of an nSt matrix *need not* be orthogonal. We recall from [15] the notion of *compact Stiefel manifold*, denoted by $St(k,n) = \{X \in \mathbb{R}^{n \times k} \mid X^{\top}X = I_k\}.$

Lemma IV.1 (Characterization of nSt matrices). The set nSt(n, k), $k \le n$, is a compact and analytic submanifold of $\mathbb{R}^{n \times k}$ of dimension (k - 1)n + 1 - k(k + 1)/2, and it is also a submanifold of the compact Stiefel manifold (and thus, $nSt(n, k) \subseteq St(k, n)$). Moreover,

(i) nSt(n,n) is the set of orthogonal matrices, (ii) for k = 1, the matrix V is nSt if and only if

$$V = \frac{1}{\sqrt{n}} \begin{bmatrix} s_1 \\ \vdots \\ s_n \end{bmatrix}, \tag{15}$$

for any numbers $s_i \in \{-1, +1\}$, $i \in \{1, ..., n\}$, (iii) for k = 2, the matrix V is nSt if and only if

$$V = \sqrt{\frac{2}{n}} \begin{bmatrix} \cos \alpha_1 & \sin \alpha_1 \\ \vdots & \vdots \\ \cos \alpha_n & \sin \alpha_n \end{bmatrix}, \quad (16)$$

for any set of angles $\alpha_1, \ldots, \alpha_n$ satisfying

$$\sum_{n=1}^{n} e^{2\alpha_m \sqrt{-1}} = 0.$$
 (17)

We postpone the proof of Lemma IV.1 to Appendix A. We remark that in the case of n = k = 2, the constraint (17) implies that $2\alpha_2 = \pi + 2\pi s + 2\alpha_1$, where $s \in \mathbb{Z}$, that is, $\alpha_2 = \pi/2 + \pi s + \alpha_1$ and $\cos \alpha_2 = (-1)^{s+1} \sin \alpha_1$, $\sin \alpha_2 = (-1)^s \cos \alpha_1$. Thus, the matrices in nSt(2, 2) are orthogonal 2×2 matrices (representing proper or improper rotations):

$$V = \begin{bmatrix} \cos \alpha_1 & \sin \alpha_1 \\ -\varepsilon \sin \alpha_1 & \varepsilon \cos \alpha_1 \end{bmatrix}, \quad \varepsilon \in \{-1, +1\}.$$

For a general k, it is difficult to give a closed-form description of all matrices from nSt(n, k). However, there are simple examples of matrices from nSt(n, k) in the case where n = 2k, including every matrix of the form

$$V = \frac{1}{\sqrt{2}} \begin{bmatrix} U_1 \\ U_2 \end{bmatrix},$$

where U_i are orthogonal $k \times k$ matrices.

B. Technical results

The classification of equilibria relies on the following technical results that will be proved in Appendix A.

Lemma IV.2. Suppose that $Z^2 - 2\alpha Z = \beta I_n$ for some symmetric $n \times n$ matrix Z with $\operatorname{diag}(Z) = \mathbb{O}_{n \times n}$ and scalars α, β . Then Z can be decomposed as

$$Z = pVV^{\top} - qI_n = Z^{\top} \tag{18}$$

Corollary IV.3. Given a matrix $Z = Z^{\top}$ with diag $(Z) = \mathbb{O}_{n \times n}$, the matrix $Z^2 - 2\alpha Z$ is diagonal with s different eigenvalues $\beta_1 < \ldots < \beta_s$ of multiplicities n_1, \ldots, n_s respectively $(n_1 + n_2 + \ldots + n_s = n)$ if and only if there exists such a permutation matrix S that

$$SZS^{-1} = \operatorname{diag}(Z_1, \ldots, Z_s),$$

where each Z_i is decomposed as (18) with parameters p_i, q_i, V_i , where $V_i \in \mathbf{nSt}(n_i, k_i)$ for some $k_i < n_i$ and

$$p_i = 2\sqrt{\alpha^2 + \beta_i}, \quad q_i = \sqrt{\alpha^2 + \beta_i} - \alpha.$$
 (19)

Thus, for irreducible $Z = Z^{\top}$ the matrix $Z^2 - 2\alpha Z$ is diagonal if and only if Z is decomposed as (18) with $p, q \ge 0$.

C. Classification of irreducible symmetric equilibria

Theorem IV.4 (Irreducible equilibria for the projected pure-influence model). *For the projected pure-influence model* (9), (*i*) all irreducible symmetric equilibria are of the form

$$Z^* = pVV^{\top} - qI_n, \qquad (20)$$

with $V \in \mathbf{nSt}(n,k)$, $1 \le k < n$, and

$$p = \sqrt{\frac{n}{k(n-k)}}, \quad q = \sqrt{\frac{k}{n(n-k)}}; \qquad (21)$$

- (ii) Z^* has k positive eigenvalues with value p-q and n-knegative eigenvalues with value -q;
- (iii) the dissonance function satisfies

$$\mathcal{D}(Z^*) = -\frac{n-2k}{\sqrt{kn(n-k)}},\tag{22}$$

and the right-hand side is monotonically increasing in $k \in \{1, ..., n-1\}$ (see Figure 2).



Fig. 2. For a network with size n = 10, the dissonance function \mathcal{D} evaluated on all irreducible symmetric equilibria with $k \in \{1, \ldots, 9\}$ positive eigenvalues, according to equation (22).

Proof. We start by proving a technical statement. Pick $V \in nSt(n,k)$, p,q real numbers and set $\theta = p - 2q$. Then, the matrix $Z = pVV^{\top} - qI_n = Z^{\top}$ satisfies the following properties:

- (a) $Z^2 \theta Z = q(p-q)I_n$, and thus diag $(Z^2) = \theta$ diag $(Z) + q(p-q)I_n$;
- (b) for any $p \neq 0$, the matrix Z has two eigenvalues p-q and (-q) whose multiplicities are k and (n-k) respectively;

- (c) the eigenspaces corresponding to p q and -q are the image of V and the kernel of V^{\top} respectively;
- (d) diag $(Z) = \mathbb{O}_{n \times n}$ if and only if pk = qn; in this situation, trace $(Z^2) = q(p-q)n$ and $\mathcal{D}(Z) = -\operatorname{trace}(Z^2 Z^{\top}) = -\theta nq(p-q).$

To prove (a), recall that $V^{\top}V = I_k$ and therefore

$$Z^2 = p^2 V V^\top V V^\top + q^2 I_n - 2pq V V^\top = p\theta V V^\top + q^2 I_n$$
$$= \theta Z + (pq - q^2) I_n.$$

To prove (b) and (c), notice that for any vector z = Vy one has $VV^{\top}z = V(V^{\top}V)y = Vy = z$, and thus Zz = (p-q)z. The space of such vectors is nothing else than the image of V and has dimension k (recall that the columns of V are orthogonal, and hence are linearly independent). If $V^{\top}z =$ 0, then Zz = -qz, and the dimension of ker (V^{\top}) is (n - k). Since $Z = Z^{\top}$ and $p - q \neq -q$ (except for the case where p = q = 0 and Z = 0, which is trivial), the two eigenspaces are orthogonal and their sum coincides with \mathbb{R}^n . Hence, there are no other eigenvalues. To prove (d), note first $p \operatorname{diag}(VV^{\top}) = (pk/n)I_n$, and thus $\operatorname{diag}(Z) = \mathbb{O}_{n \times n}$ if and only if pk/n = q. Using statement (a), one shows that in this situation $\operatorname{diag}(Z^2) = q(p-q)I_n$ and hence $\operatorname{trace}(Z^2) = q(p-q)n$. Thanks to (a), $Z^3 = \theta Z^2 + q(p-q)Z \Longrightarrow \operatorname{trace}(Z^3) =$ $\theta \operatorname{trace}(Z^2) = \theta nq(p-q)$, which finishes the proof of (d).

Now, to prove the statement (i) of the theorem, let Z^* be an irreducible symmetric solution to equation (14). For $\alpha = -\mathcal{D}(Z^*)/2$, the matrix $(Z^*)^2 - 2\alpha Z^* = \text{diag}(Z^{*2})$ is diagonal. Since Z^* is irreducible, it follows from Corollary IV.3 that Z^* can be decomposed as (20) with some $p, q \ge 0$. Then, from (a) and (d), it also follows that Z^* satisfies equation (14) if and only if pk = qn (which comes from $\text{diag}(Z^*) = \mathbb{O}_{n \times n}$) and $pq - q^2 = 1/n$ (which comes from $\text{trace}(Z^{*2}) = 1$). This implies that $q = \sqrt{\frac{k}{n(n-k)}}$ and $p = \sqrt{\frac{n}{k(n-k)}}$.

Finally, statement (ii) follows from (b); and (iii) is obtained by substituting the values of p and q into the definition of the dissonance function (5) and noting that the smooth function $\kappa \mapsto -\frac{n-2\kappa}{\sqrt{n\kappa(n-\kappa)}}$ has positive derivative on (0, n).

D. Classification of reducible symmetric equilibria

The next theorem generalizes Theorem IV.4 and characterizes all symmetric equilibria for the projected pure-influence model and its proof can be found in Appendix A.

Theorem IV.5 (All equilibria for the projected pure-influence model). The matrix Z^* is an equilibrium (14) of the projected pure-influence model if and only if a permutation matrix S exists such that:

- (i) $SZ^*S^{-1} = \text{diag}(Z_1^*, \dots, Z_s^*), \ s \ge 1, \ Z_i^* = Z_i^{*\top} \in \mathbb{R}^{n_i \times n_i}$
- (ii) the blocks Z_i^* admit representation (18): $Z_i^* = p_i V_i V_i^\top q_i I_{n_i}$, where $p_i, q_i \ge 0$ and $V_i \in \operatorname{nSt}(n_i, k_i)$, $1 \le k_i < n_i$;
- (iii) the sign $\varepsilon = \operatorname{sign}(n_i 2k_i) \in \{-1, 0, 1\}$ is the same for all $i = 1, \ldots, s$ such that $Z_i^* \neq \mathbb{O}_{n_i \times n_i}$ and
- (iv) each block $Z_i^* \neq \mathbb{O}_{n_i \times n_i}$ is irreducible and the corresponding coefficients p_i, q_i have the form

$$p_i = 2\sqrt{\alpha^2 + \beta_i}, \quad q_i = \sqrt{\alpha^2 + \beta_i} - \alpha,$$
 (23)

where

a) for $\varepsilon \neq 0$, α and β_i are determined from

$$\alpha = \varepsilon \left(\sum_{i:Z_i \neq 0} \frac{4k_i n_i (n_i - k_i)}{(n_i - 2k_i)^2} \right)^{-1/2},$$

$$\beta_i = \alpha^2 \frac{4n_i k_i - 4k_i^2}{(n_i - 2k_i)^2};$$
(24)

b) for ε = 0, α = 0, for all i, and β_i are chosen in such a way that Σ_{i:Zi≠0} β_in_i = 1.

Remark IV.6. Let Z^* be a reducible equilibrium for the projected pure-influence model such that $G(Z^*)$ is composed of m (disconnected) subgraphs that satisfy structural balance. According to Definition II.3, $G(Z^*)$ does not satisfy structural balance since this definition requires $G(Z^*)$ to be complete.

E. Structural balance and equilibria

We now characterize the equilibria corresponding to structural balance and how they minimize the dissonance function.

Corollary IV.7 (Balanced equilibria of the projected pure-influence model). For the projected pure-influence model (9), let $Z^* \in \mathbb{S}_{\text{zero-diag}}^{n \times n}$ be an equilibrium point with a single positive eigenvalue. Then,

(i) after a relabelling of the agents, Z^* has the form

$$Z^* = \begin{bmatrix} Z' & \mathbb{O}_{n_1 \times (n-n_1)} \\ \hline \mathbb{O}_{(n-n_1) \times n_1} & \mathbb{O}_{(n-n_1) \times (n-n_1)} \end{bmatrix}$$
(25)

with $n_1 \leq n$ and

$$Z' = \frac{1}{\sqrt{n_1(n_1 - 1)}} (ss^{\top} - I_{n_1}), \qquad (26)$$

for some $s \in \{-1, +1\}^{n_1}$; and thus, for any fixed n_1 , there are only 2^{n_1-1} different equilibria (with a single positive eigenvalue),

- (ii) G(Z') satisfies structural balance, with the binary vector s characterizing the distribution of the individuals in the single faction or in the two factions, and
- (iii) if $G(Z^*)$ is a connected graph, then $G(Z^*)$ satisfies structural balance (being thus complete) and Z^* is a global minimizer to the optimization problem:

$$\begin{array}{ll} \underset{Z \in \mathbb{R}^{n \times n}}{\text{minimize}} & \mathcal{D}(Z) \\ \text{subject to} & Z \in \mathbb{S}^{n \times n}_{\text{zero-diag,symm}} \\ \text{and satisfies } \mathcal{D}(Z^*) = -\frac{n-2}{\sqrt{n(n-1)}}. \end{array}$$

Proof. Consider a permutation of indices from Theorem IV.5. Since Z^* has only one positive eigenvalue, it can have only one non-zero diagonal block $Z_i^* = Z'$. Statement (i) now follows from (20),(21) (with $k = 1, n = n_1$) and (15).

Regarding statement (ii), observe that for any different i, j and k,

$$z_{ij}' z_{jk}' z_{ki}' = \frac{(s_i s_j)(s_j s_k)(s_k s_i)}{(n_1(n_1 - 1))^{3/2}} = \frac{1}{(n_1(n_1 - 1))^{3/2}} > 0.$$

This inequality implies $sign(z'_{ij}) = sign(z'_{jk}z'_{ki})$ and thus we know that Z' satisfies structural balance. It is immediate to

see that any i and j such that $s_i = s_j$ correspond to the same faction in the network G(Z'). This completes the proof for (ii).

Regarding statement (iii), we notice that the smooth function $\eta \mapsto -\frac{\eta-2}{\sqrt{\eta(\eta-1)}}$ has negative derivative for $\eta > 3/2$. Hence, the value of $\mathcal{D}(Z^*) = \mathcal{D}(Z') = -\frac{n_1-2}{\sqrt{n_1(n_1-1)}}$ at equilibrium (25) with one positive eigenvalue is minimal when $Z' = Z^*$ and $n_1 = n$, that is, the matrix is irreducible. Now, let us focus on the points that vanish the gradient of \mathcal{D} , i.e., the equilibria of the projected pure-influence model. Permuting the agents, we may confine ourselves to equilibria described in Theorem IV.5 that have s blocks of size n_i with $k_i < n_i$ positive eigenvalues, $i \in \{1, \ldots, s\}$. To see why this is true, in the proof of Theorem IV.4 it was shown that $\mathcal{D}(Z_i^*) = -2\alpha n q_i (p_i - q_i) = -2\alpha \beta_i$. Next, if $\varepsilon = -1$, then $\alpha < 0$ and $D(Z^*) > 0$. If $\varepsilon = \text{sign}(n_i - 2k_i) = 0$ for all $Z_i^* \neq 0$, then $\mathcal{D}(Z^*) = \sum_i \mathcal{D}(Z_i^*) = 0$. As we know, the minimal value should be negative, so such equilibria cannot be global minimizers. Therefore, we may assume that $\varepsilon = 1$, that is, $k_i < n_i/2$ for all such i that $Z_i^* \neq 0$. Assume, without loss of generality, that $Z_1^*, \ldots, Z_m^* \neq 0$ and $Z_{m+1}^*, \ldots, Z_s^* = 0$. Denote $k_1 + \cdots + k_m = k'$ and $n_1 + \cdots + n_m = n' \leq n$. Note that the function $f(\xi) = \xi(1-\xi)/(1-2\xi)^2$ is convex on (0, 1/2). Therefore, Jensen's inequality implies

$$\frac{1}{n} \sum_{i=1}^{m} \frac{k_i n_i (n_i - k_i)}{(n_i - 2k_i)^2} = \sum_{i=1}^{m} \frac{n_i}{n} f\left(\frac{k_i}{n_i}\right)$$
$$\geq f\left(\sum_{i=1}^{m} \frac{k_i}{n'}\right) = f\left(\frac{k'}{n'}\right) = \frac{k'(n' - k')}{(n' - 2k')^2},$$

and, in turn,

$$\mathcal{D}(Z^*) = -\left(\sum_{i=1}^m \frac{k_i n_i (n_i - k_i)}{(n_i - 2k_i)^2}\right)^{-1/2} \ge -\frac{n' - 2k'}{\sqrt{k'n'(n' - k')}}$$

We know, however from Theorem IV.4 that the right-hand side is minimal when k' = 1, in which case the minimal value, as we have seen in the beginning in the proof, is achieved at n' = n. Hence, the irreducible equilibrium with one positive eigenvalue is the global minimizer of \mathcal{D}^* .

Remark IV.8. Let Z^* denote an equilibrium point with one positive eigenvalue. Then, $-Z^*$ has one negative eigenvalue and does not correspond to structural balance. All such $-Z^*$ correspond to isolated critical points of \mathcal{D} .

F. Examples of equilibria with two positive eigenvalues

Let Z^* be any equilibrium of the projected pure-influence model parameterized by nSt(n, 2) matrices, so that it has two positive eigenvalues. Let us assume first that it is irreducible. Then, another class of equilibria is found using the parametrization (16). It can be easily shown that

$$Z^* = \sqrt{\frac{2}{n(n-2)}} (\theta_{ij})_{i,j=1}^n, \quad \theta_{ij} = \begin{cases} 0, \ i=j\\ \cos(\alpha_i - \alpha_j), \ i \neq j. \end{cases}$$

Here the angles α_i should satisfy the relation (17). Interestingly, many of such matrices do not correspond to structural balance. Consider, for example, the case where the unit vectors in (17) constitute a regular *n*-gon: $\alpha_i = \frac{\pi(i-1)}{n}$, i = 1, ..., n. For any pair i, j > i the entry z_{ij} is negative if (j-i) > n/2, positive if j - i < n/2 and zero if j - i = n/2 (possible only for even *n*). If *n* is odd, the graph is complete, otherwise, the pairs of nodes (i, i + n/2) for i = 1, ..., n/2 are not connected. For example, in the smallest dimension n = 3, by setting $\alpha_1 = 0$, $\alpha_2 = \pi/3$ and $2\pi/3$, we obtain the equilibrium

$$Z^* = \frac{1}{\sqrt{6}} \begin{bmatrix} 0 & +1 & -1\\ +1 & 0 & +1\\ -1 & +1 & 0 \end{bmatrix}$$

which does not correspond to structural balance. Actually, in the case where n = 3 or $n \ge 5$, the graph always contains imbalanced triads. For instance, for $n \ge 3$ being odd the nodes i = 1, j = (n - 1)/2 and $\ell = (n + 3)/2$ always constitute such a triad: $z_{i\ell} < 0$, whereas $z_{ij}, z_{j\ell} \ge 0$. For an even number $n \ge 6$, one may take $i = 1, j = n/2, \ell = n/2 + 2$. In the case n = 4, the equilibrium Z^* corresponds to an incomplete cyclic graph such that $\mathcal{D}(Z^*) = 0$:

$$Z^* = \frac{1}{2\sqrt{2}} \begin{bmatrix} 0 & \frac{1}{\sqrt{2}} & 0 & -\frac{1}{\sqrt{2}} \\ \frac{1}{\sqrt{2}} & 0 & \frac{1}{\sqrt{2}} & 0 \\ 0 & \frac{1}{\sqrt{2}} & 0 & \frac{1}{\sqrt{2}} \\ -\frac{1}{\sqrt{2}} & 0 & \frac{1}{\sqrt{2}} & 0 \end{bmatrix}$$

For the reducible matrix case, since Z^* has two positive eigenvalues, $G(Z^*)$ contains two disconnected subgraphs that satisfy structural balance with possibly other isolated nodes.

V. CONVERGENCE TO BALANCED EQUILIBRIA AND STABILITY ANALYSIS

We now provide convergence results for our models towards equilibria that correspond to structural balance. We present a supporting lemma and then our main theorem.

Lemma V.1. Assume that the solution of (2) satisfies $x_{i*}(t_0) = \mathbb{O}_{1 \times n}$ at some $t_0 \ge 0$, that is, in the graph $G(X(t_0))$ node *i* does not communicate to any other node. Then, $x_{i*}(t) \equiv \mathbb{O}_{1 \times n}$ for any $t \ge 0$. The same holds for the solutions of (9).

Proof. Since the right-hand sides of (2) and (9) are analytic, any solution is a real-analytic function of time. Assuming that $x_{ij}(t_0) = 0$ for all j, one finds that $\dot{x}_{ij}(t_0) = 0$. Differentiating (2), it is easy to show that $\ddot{x}_{ij}(t_0) = 0$, and so on, $x_{ij}^{(m)}(t_0) = 0$ for any $m \ge 1$. In view of analyticity, one has $x_{ij}(t) \equiv 0$ for any t. Similarly, $z_{ij}(t_0) = 0 \forall j$ entails that $z_{ij}(t) \equiv 0$ for any solution of (9).

Theorem V.2 (Convergence results and dynamical properties). Consider the pure-influence model (2) with an initial condition $X(0) \in \mathbb{R}^{n \times n}_{\text{zero-diag,symm}}$ and the projected pure-influence model (9) with initial condition $Z(0) = \frac{X(0)}{\|X(0)\|_F}$. Then,

- (i) the solution Z(t) converges to a single critical point of the dissonance function D;
- (ii) the number of negative eigenvalues of Z(t) is nondecreasing.
- Moreover, if X(0) has one positive eigenvalue, then

- (iii) $\lim_{t\to+\infty} Z(t) = Z^*$, where Z^* is as in (26), so that G(Z(t)) or one of its connected components (while the rest of nodes are isolated) reaches structural balance in finite time;
- (iv) X(t) achieves the same sign structure as Z^* in finite time;
- (v) nonzero entries of X(t) diverge to infinity in finite time.

Proof. For convenience, throughout this proof, let us denote $W(t) = \frac{X(t)}{\|X(t)\|_F}$, i.e., $X(t) = \eta(t)W(t)$ with $\eta(t)$ evolving according to (6a) and W(t) evolving according to (6b). From the construction of the transcription of the pure-influence model in Theorem III.3, we have that $\eta(t) = \|X(t)\|_F$ and so $\eta(t) > 0$ for all well-defined $t \ge 0$. Moreover, Lemma A.1 let us conclude that $W(t) = Z(\int_0^t \eta(s) ds)$ for all $t \ge 0$, and thus the solution X(t) is well defined.

To prove (i), recall that (9) is a gradient flow dynamics of the analytic function \mathcal{D} , and the trajectory Z(t) stays on a compact manifold and, in particular, is bounded. The classical result of Łojasiewicz [1] implies convergence of the trajectory to a single fixed point.

To prove (ii), we enumerate the eigenvalues of Z(t) in the descending order $\lambda_1(t) \geq \lambda_2(t) \dots \geq \lambda_n(t)$ and consider the corresponding orthonormal bases of eigenvectors $v_i(t)$. Since $Z_i(t)v_i(t) = \lambda_i(t)v_i(t)$ and $v_i(t)^{\top}v_i(t) = 1$, we obtain $\dot{Z}v_i + Z\dot{v}_i = \dot{\lambda}_i v_i + \lambda_i \dot{v}_i$ and $\dot{v}_i(t)^{\top}v_i(t) = 0$. Therefore,

$$\dot{\lambda}_i = v_i^\top \dot{Z} v_i + v_i^\top Z \dot{v}_i = v_i^\top \dot{Z} v_i + \lambda_i v_i^\top \dot{v}_i = v_i^\top \dot{Z} v_i,$$

entailing the following differential equation

$$\dot{\lambda}_i = \lambda_i^2 + \mathcal{D}(Z)\lambda_i - v_i^\top \operatorname{diag}(Z^2)v_i.$$
(27)

Notice that all diagonal entries of $\operatorname{diag}(Z^2)$ are nonnegative. Now, due to Lemma V.1, if the *i*th row of X was initially the zero vector, then it will continue being the same for all times and also for Z; and, moreover, $\operatorname{diag}(Z^2)_{ii} = 0$ and there exists a zero eigenvalue with its associated eigenvector having zero entries in all the positions of the entries where $\operatorname{diag}(Z^2)$ are positive. Then, it immediately follows from (27) that if $\lambda_i(0) = 0$ due to Z(0) having a row being the zero vector $\mathbb{O}_{1 \times n}$, then $\dot{\lambda}_i = 0$.

Now, let \mathcal{N} be the set of indices i such that $\operatorname{diag}(Z^2)_{ii} > 0$. Thus, for any $i \in \mathcal{N}$, if λ_i crosses the real axis at time t^* , i.e., $\lambda(t^*) = 0$, then

$$\dot{\lambda}_i(t^*) = -(v_i(t^*))^\top \operatorname{diag}\left(Z^2(t^*)\right)v_i(t^*) < 0.$$
 (28)

Therefore, if $\lambda_i(t_0) \leq 0$ for some $t_0 \geq 0$, then $\lambda_i(t) \leq 0$ for all $t \geq t_0$. This finishes the proof for (ii).

Notice that since $\operatorname{trace}(Z(t)) = 0$ and $Z(t) = Z(t)^{\top} \neq \mathbb{O}_{n \times n}$, then Z(t) has at least one positive eigenvalue. Then, equation (28) implies that

 $\Lambda := \{ Z \in \mathbb{S}^{n \times n}_{\text{zero-diag,symm}} \mid Z \text{ has only one positive eigenvalue} \}$

is forward invariant and, in particular, the limit $Z^* = \lim_{t\to\infty} Z(t)$ (existing in view of statement (i)) belongs to Λ . Since Z^* is a critical point of \mathcal{D} (or, in view of Theorem III.4, the equilibrium of (9)), it has the structure described by Corollary IV.7. By continuity of the flow Z(t), there is a finite time τ such that G(Z(t)) has the same sign structure as $G(Z^*)$ for all $t \geq \tau$. This finishes the proof for (iii).

Now we prove the last two statements of the theorem. Knowing the convergence result from (iii), Lemma A.1 tells us that introducing the term η as in the transcribed system (6a) to the projected pure-influence model has the simple effect of altering the convergence rate properties for Z(t). Therefore, there always exist a finite time $\tau^* \geq 0$ such that, for any $t \geq \tau^*$, W(t) satisfies the sign properties of statement (iii) regarding structural balance. Moreover, the fact that $X(t) = \eta(t)W(t)$ and $\eta(t) \ge 0$ by construction, immediately implies (iv). Now, let $q(t) := -\mathcal{D}(W(t))$, and notice that q(t) is a strictly positive continuous function for all (well-defined) $t \ge \tau^*$. Now, from equation (6b), we have the system $\dot{\eta}(t) = g(t)\eta^2(t)$, with solution $\eta(t) = \frac{\eta(\tau)}{1 - \eta(\tau) \int_{\tau}^{t} g(s)ds}$ for $t \geq \tau$. Then, since $\int_{\tau}^{t} g(s) ds$ is a monotonic strictly increasing function on $t \geq \tau$, we have that $\eta(t) \rightarrow +\infty$ as $t \to t^*$, where $t^* > \tau^*$ is some finite time such that $\int_{\tau}^{t^*} g(s)ds = \frac{1}{\eta(\tau)}$ (note that $t^* > \tau^*$ holds from the relationship $W(t) = Z(\int_0^t \eta(s)ds)$). Then, we conclude that the solution $\eta(t)$ and the entries of X(t) diverge in some finite time t^* , which proves (v).

Corollary V.3. Consider the same conditions as in Theorem V.2, i.e., the projected pure-influence model with initial condition $Z(0) \in \mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$ having one positive eigenvalue. If $\mathcal{D}(Z(0)) < -\frac{n-3}{\sqrt{(n-1)(n-2)}}$, then G(Z(t)) eventually reaches structural balance.

The previous theorem immediately implies that the set of irreducible equilibria with a single positive eigenvalue is (locally) asymptotically stable. We present further results on the stability of equilibria.

Lemma V.4 (Further results on stability of the equilibria). Consider a symmetric equilibrium point Z^* for the projected pure-influence model (9). Without loss of generality, assume that Z^* has no row equal to the zero vector¹. If $\mathcal{D}(Z^*) \ge 0$, then Z^* is an unstable equilibrium point and does not correspond to structural balance.

Proof. Write the analytic projected influence system (9) as $\dot{Z} = f(Z) := Z^2 - \text{diag}(Z^2) + \mathcal{D}(Z)Z$, thereby defining $f : \mathbb{R}^{n \times n} \to \mathbb{R}^{n \times n}$, and compute

$$\frac{\partial f_{ij}(Z)}{\partial z_{ij}} = \mathcal{D}(Z) + \frac{\partial \mathcal{D}(Z)}{\partial z_{ij}} z_{ij},$$
$$\frac{\partial \mathcal{D}(Z^*)}{\partial z_{ij}} = -3 \sum_{\substack{k=1\\k \neq i,j}}^{n} z_{ik}^* z_{kj}^*.$$

Now, the Jacobian of f, denoted by Df, is a $(n^2-n) \times (n^2-n)$ matrix (since we do not consider self-appraisals). Let $Df(Z^*)$

be the Jacobian evaluated at Z^* and let $\{\lambda_i\}_{i=1}^{n^2-n}$ be the set of its eigenvalues. Then, we compute

$$\sum_{i=1}^{n^2 - n} \lambda_i = \text{trace}(Df(Z^*)) = \sum_{i=1}^n \sum_{\substack{j=1 \ j \neq i}}^n \frac{\partial f_{ij}(Z^*)}{\partial z_{ij}}$$
$$= (n^2 - n)\mathcal{D}(Z^*) + 3\mathcal{D}(Z^*) = (n^2 - n + 3)\mathcal{D}(Z^*).$$

Since $n^2 - n + 3 > 0$ for $n \ge 3$, we draw the following conclusions for $\mathcal{D}(Z^*) \ge 0$: (i) $Df(Z^*)$ contains at least one positive eigenvalue and so the equilibrium point Z^* is unstable; (ii) at least one triad in $G(Z^*)$ is unbalanced and so Z^* does not correspond to structural balance.

VI. SIMULATION RESULTS AND CONJECTURES

The generic convergence of trajectories to the minima of \mathcal{D} (or, equivalently, the convergence from almost all initial conditions) is an open problem. However, we present strong numerical evidence that support such claim. We first remark that, from the proof of Theorem III.3, the projected pure-influence model (9) can be generalized over any asymmetric matrix in $\mathbb{S}_{\text{zero-diag}}^{n \times n}$ by replacing $\mathcal{D}(Z)$ by $- \text{trace}(Z^{\top}Z^2)$ and this is the model we will refer throughout this section.

A generic asymmetric initial condition X(0) for the pureinfluence model (2) is a matrix that is generated with each entry independently sampled from a uniform distribution with support [-100, 100], and its diagonal entries set to zero. A generic symmetric initial condition is similarly constructed by only sampling the upper triangular entries of the matrix. For the projected pure-influence model, we say $Z(0) = \frac{X(0)}{\|X(0)\|_F}$ is a (non-)symmetric generic initial condition depending on how X(0) was generated. We immediately see from the proof of Theorem V.2, that Z(t) converges to social balance if and only if X(t) converges to social balance. Indeed, given that X(t)diverges at some finite time \bar{t} , we have $Z(\infty) = \frac{X(\bar{t}^-)}{\|X(\bar{t}^-)\|_F}$.

For a fixed network size n, we use a Monte Carlo method [26] to estimate the probability p of the event "under a generic asymmetric initial condition Z(0), Z(t) converges to structural balance in finite time". We estimate p by performing N independent simulations (i.e., each simulation generates a new independent initial condition) and obtaining the proportion \hat{p}_N , also known as the empirical probability, of times that the simulation indeed had Z(t) converging to structural balance in finite time. For any accuracy $1 - \epsilon \in (0, 1)$ and confidence level $1 - \eta \in (0, 1)$ we have that $|\hat{p}_N - p| < \epsilon$ with probability greater than $1 - \eta$ if the Chernoff bound $N \ge \frac{1}{2\epsilon^2} \log \frac{2}{\eta}$ is satisfied. For $\epsilon = \eta = 0.01$, the bound is satisfied by N = 27000. We performed the N = 27000independent simulations with $n \in \{5, 6\}$, and found that $\hat{p}_N = 1$. Our observations let us conclude that for generic asymmetric initial condition Z(0) and $n \in \{5, 6\}$, with 99% confidence level, there is at least 0.99 probability that Z(t)converges to structural balance in finite time.

Similarly, we performed the same Monte Carlo analysis for generic symmetric initial conditions with $n \in \{3, 5, 6, 15\}$, and found for that $\hat{p}_N = 1$ for all n. Therefore, we conclude that for any symmetric generic initial condition Z(0) and $n \in \{3, 5, 6, 15\}$, with 99% confidence level, there is at least 0.99

¹If Z^* had a row equal to the zero vector, then, in the lemma statement, we would replace n by $n_1 < n$, where n_1 is the number of rows of Z^* that are not equal to the zero vector.

probability that Z(t) converges to structural balance in finite time.

We report three more observations and then state a resulting conjecture. First, remarkably, we found that all of our simulations (for any type of random initial condition) that converged to structural balance in finite time, did it by converging to an equilibrium point having only one positive eigenvalue inside the set of scale-symmetric matrices, which is a superset of the set of symmetric matrices (see Appendix B). Second, we did not perform experiments for larger sizes of n due to computational constraints. Third, unfortunately, for n = 3, we did find randomly-generated asymmetric initial conditions whose numerically-computed solutions do not converge to structural balance.

Conjecture 1 (Convergence from generic initial conditions). Consider the pure-influence model (2) with some initial condition X(0), and the projected pure-influence model (9) with initial condition $Z(0) = \frac{X(0)}{\|X(0)\|_F}$. Then,

(i) under generic asymmetric initial conditions, $\lim_{t\to+\infty} Z(t) = Z^*$ for a sufficiently large n,

(ii) under generic symmetric initial conditions, $\lim_{t\to+\infty} Z(t) = Z^*$ for any n,

where Z^* is scale-symmetric (and particularly symmetric for (ii)) corresponding to structural balance. Then, Z(t)reaches structural balance in finite time. Moreover, X(t)reaches structural balance in finite time with same sign structure as Z^* , and also diverges in finite time.

Similarly, we performed the same simulation analysis for the Kułakowski et al. model (4), which converges to structural balance if and only if the projected Kułakowski model (10) does. To generate a generic initial condition for this system, we generated an $n \times n$ matrix with each entry independently sampled from a uniform distribution with support [-100, 100], and then divide it by its Frobenius norm. We performed N =27000 independent simulations with $n \in \{5, 6\}$, and found that for generic initial condition Z(0) and n = 5, only 16.94% converged to structural balance, and for n = 6, only 11.50% converged to structural balance.

Also, for n = 3, not all simulations converged to structural balance. We remark that not all of the networks for which the system converged and did not satisfy structural balance were complete, some of them were networks with only selfloops, e.g., Figure 5(a). Similarly, we performed the same Monte Carlo analysis for symmetric initial conditions with $n \in \{3, 5, 6, 15\}$. Our results show that for symmetric generic initial condition, Z(0) did not always converge to structural balance for n = 3, but, for $n \in \{5, 6, 15\}$, with 99% confidence level, there is at least 0.99 probability that Z(t)converges to structural balance in finite time.

These Monte Carlo results are expected, since it has been formally proved that the Kułakowski et al. model converges to structural balance only under generic symmetric initial conditions as $n \to \infty$ [22] and negative results for asymmetric conditions are given by [27].

See Figure 3 for a comparison of trajectories of the pureinfluence model in both generic and symmetric generic initial conditions. Figure 4 shows a comparison between our projected pure-influence model, which does not consider selfappraisals, and the projected influence model, which considers self-appraisals. Note how not considering self-appraisals drastically changes the convergence time as well as the dynamic behavior of the interpersonal appraisals.



(a) Projected pure-influence model (9) (b) Projected pure-influence with generic asymmetric initial condi- model (9) with generic symmetric tion initial condition

Fig. 3. Convergence to structural balance for a network of size n = 10. We plot the evolution of all the entries of Z(t).



Fig. 4. Convergence comparison for a network of size n = 7 (a) with and (b) without the consideration of self-appraisals. We first generated an $n \times n$ random matrix W with each entry independently sampled from a uniform distribution with support [-100, 100]. Then, for (a), we normalize this matrix to have unit Frobenius norm and used it as the initial condition. For (b), we set the diagonal entries of W to zero and then normalize it to have unit Frobenius norm and use it as the initial condition. In this example, (a) did not converged to structural balance, whereas (b) did. We plot the evolution of all the entries of the appraisal matrix.



Fig. 5. Convergence comparison for a network of size n = 7 (a) with and (b) without the consideration of self-appraisals. The setting is the same one as in Figure 4, but with a different random initial condition. (a) converged to a network with only diagonal negative entries (all interpersonal appraisals go to zero), whereas (b) converged to structural balance.

VII. CONCLUSION

We propose two new dynamic structural balance models that incorporates more psychologically plausible assumptions than previous models in the literature, based on a modification by a model proposed by Kułakowski et al. We have established important convergence properties for these models and also that, most importantly, they correspond to gradient systems over an energy function that characterizes the violations of Heider's axioms for the symmetric case. We also expanded our results to a set of asymmetric matrices called scalesymmetric. Numerical results illustrates that, under generic initial conditions, our models converges to structural balance (for sufficiently large n) and thus have better convergence properties than the previous model by Kułakowski et al.

As future work, we propose to further study the general case of asymmetric (and non-scale-symmetric) equilibria and the convergence properties of our models under arbitrary initial conditions. For example, numerical simulations of the projected pure-influence model from generic (asymmetric) initial conditions illustrate how this system features transient chaos before converging towards an equilibrium. Future work will focus on models with a more sociologically justified transient behavior. Finally, one could study the removal of the self-appraisals in other dynamical structural balance models, like the homophily-based model by Traag et al. [27].

ACKNOWLEDGMENT

We are grateful to Prof. John Gilbert, Prof. Ambuj Singh, Dr. Saber Jafarpour, and Prof. Maria Elena Valcher for insightful discussions and helpful feedback.

REFERENCES

- P.-A. Absil, R. Mahony, and B. Andrews. Convergence of the iterates of descent methods for analytic cost functions. *SIAM Journal on Control* and Optimization, 6(2):531–547, 2005. doi:10.1137/040605266.
- [2] T. Antal, P. L. Krapivsky, and S. Redner. Dynamics of social balance on networks. *Physical Review E*, 72(3):036121, 2005. doi:10.1103/ PhysRevE.72.036121.
- [3] T. Antal, P. L. Krapivsky, and S. Redner. Social balance on networks: The dynamics of friendship and enmity. *Physica D: Nonlinear Phenomena*, 224(1):130–136, 2006. doi:10.1016/j.physd.2006. 09.028.
- [4] D. Cartwright and F. Harary. Structural balance: A generalization of Heider's theory. *Psychological Review*, 63(5):277, 1956. doi:10. 1037/h0046049.
- [5] P. Cisneros-Velarde and F. Bullo. Signed network formation games and clustering balance. *Dynamic Games and Applications*, 2020. doi: 10.1007/s13235-019-00346-8.
- [6] D. Easley and J. Kleinberg. Networks, Crowds, and Markets: Reasoning About a Highly Connected World. Cambridge University Press, 2010, ISBN 0521195330.
- [7] G. Facchetti, G. Iacono, and C. Altafini. Computing global structural balance in large-scale signed social networks. *Proceedings of the National Academy of Sciences*, 108(52):20953–20958, 2011. doi: 10.1073/pnas.1109521108.
- [8] L. Festinger. A Theory of Cognitive Dissonance. Stanford University Press, 1957, ISBN 9780804709118.
- [9] N. E. Friedkin. A Structural Theory of Social Influence. Cambridge University Press, 1998, ISBN 9780521454827.
- [10] N. E. Friedkin, A. V. Proskurnikov, and F. Bullo. Positive contagion and the macrostructures of generalized balance. *Network Science*, 7(4):445– 458, 2019. doi:10.1017/nws.2019.19.
- [11] F. Harary. On the notion of balance of a signed graph. Michigan Mathematical Journal, 2(2):143-146, 1953. doi:10.1307/mmj/ 1028989917.

- F. Heider. Attitudes and cognitive organization. *The Journal of Psychology*, 21(1):107–112, 1946. doi:10.1080/00223980.1946. 9917275.
- [13] U. Helmke and J. B. Moore. Optimization and Dynamical Systems. Springer, 2 edition, 1996, ISBN 0387198571.
- [14] M. O. Jackson and S. Nei. Networks of military alliances, wars, and international trade. *Proceedings of the National Academy of Sciences*, 112(50):15277–15284, 2015. doi:10.1073/pnas.1520970112.
- [15] I. M. James and N. J. Hitchin. *The Topology of Stiefel Manifolds*. Cambridge University Press, 1976, ISBN 9780521213349.
- [16] P. Jia, N. E. Friedkin, and F. Bullo. The coevolution of appraisal and influence networks leads to structural balance. *IEEE Transactions* on Network Science and Engineering, 3(4):286–298, 2016. doi: 10.1109/TNSE.2016.2600058.
- [17] K. Kułakowski, P. Gawroński, and P. Gronek. The Heider balance: A continuous approach. *International Journal of Modern Physics C*, 16(05):707–716, 2005. doi:10.1142/S012918310500742X.
- [18] J. M. Lee. Introduction to Smooth Manifolds. Springer, 2003, ISBN 0387954481.
- [19] J. Leskovec, D. Huttenlocher, and J. Kleinberg. Signed networks in social media. In *Int. Conf. on Human Factors in Computing Systems*, pages 1361–1370, Atlanta, USA, 2010. doi:10.1145/1753326. 1753532.
- [20] C.-C. Lin, C.-H. Lee, C.-S. Fuh, H.-F. Juan, and H.-C. Huang. Link clustering reveals structural characteristics and biological contexts in signed molecular networks. *PLoS One*, 8(6):1–9, 06 2013. doi:10. 1371/journal.pone.0067089.
- [21] M. Malekzadeh, M. Fazli, P. Jalaly Khalidabadi, H. R. Rabiee, and M. A. Safari. Social balance and signed network formation games. In *Proceedings of 5th KDD Workshop on Social Network Analysis (SNA-KDD)*, San Diego, USA, August 2011.
- [22] S. A. Marvel, J. Kleinberg, R. D. Kleinberg, and S. H. Strogatz. Continuous-time model of structural balance. *Proceedings of the National Academy of Sciences*, 108(5):1771–1776, 2011. doi:10. 1073/pnas.1013213108.
- [23] S. A. Marvel, S. H. Strogatz, and J. M. Kleinberg. Energy landscape of social balance. *Physical Review Letters*, 103:198701, 2009. doi: 10.1103/PhysRevLett.103.198701.
- [24] W. Mei, P. Cisneros-Velarde, G. Chen, N. E. Friedkin, and F. Bullo. Dynamic social balance and convergent appraisals via homophily and influence mechanisms. *Automatica*, 110:108580, 2019. doi:10. 1016/j.automatica.2019.108580.
- [25] F. Radicchi, D. Vilone, S. Yoon, and H. Meyer-Ortmanns. Social balance as a satisfiability problem of computer science. *Physical Review E*, 75:026106, 2007. doi:10.1103/PhysRevE.75.026106.
- [26] R. Tempo, G. Calafiore, and F. Dabbene. Randomized Algorithms for Analysis and Control of Uncertain Systems. Springer, 2005, ISBN 1-85233-524-6.
- [27] V. A. Traag, P. Van Dooren, and P. De Leenheer. Dynamical models explaining social balance and evolution of cooperation. *PLoS One*, 8(4):e60063, 2013. doi:10.1371/journal.pone.0060063.
- [28] A. van de Rijt. The micro-macro link for the theory of structural balance. Journal of Mathematical Sociology, 35(1-3):94–113, 2011. doi:10. 1080/0022250X.2010.532262.
- [29] X. Zheng, D. Zeng, and F.-Y. Wang. Social balance in signed networks. *Information Systems Frontiers*, 17(5):1077–1095, 2015. doi: 10.1007/s10796-014-9483-8.

Appendix A

SUPPORTING RESULTS AND PROOFS

Lemma A.1. Let x(t) be the solution to $\dot{x} = f(x)$ from initial condition x(0), with f being a continuously differentiable vector field. Let η be a positive continuous scalar function. Then, z(t) is the solution to $\dot{z} = \eta(t)f(z)$ with initial condition z(0) = x(0) if and only if $z(t) = x(\int_0^t \eta(s)ds)$.

Proof. Consider the time transformation $\bar{t}(t) = \int_0^t \eta(s) ds$, which is well-defined since it is continuous and monotonically increasing on t (recall that $\eta(s) > 0$ for $s \in [0, t]$), with $\bar{t} = 0$ if and only if t = 0. Now, from the chain rule, it follows that

$$\frac{dz}{dt} = \frac{dx(t)}{d\bar{t}}\frac{dt}{dt} = f(z)\eta(t), \quad z(0) = x(0)$$

13

This finishes proof of the "if" part. The "only if" part follows from the uniqueness theorem. $\hfill\square$

Proof of Lemma IV.1. First, to prove that the set nSt(n, k), $k \leq n$ is a submanifold of the compact Stiefel manifold, define the smooth map $\Phi : St(n, k) \to \mathbb{R}^n$ by $X \mapsto$ $(||X_{i*}||_2^2, \ldots, ||X_{n*}||_2^2)^{\top}$, where X_{i*} is the *i*th row of X. Then, we have that $nSt(n, k) = \Phi^{-1}((k/n, \ldots, k/n)^{\top})$ and it is easy to prove the mapping Φ has constant rank n. Thus, we use the Constant-Rank Level Set Theorem [18] to conclude our claim. The properties of compactness and analyticity are immediate from the definition of the set nSt(n, k), $k \leq n$.

Now, notice that conditions ((i)) and ((ii)) from Definition IV.1 impose, in total, $\frac{k(k+1)}{2} + n$ constraints on kn independent variables, however, these constraints are linearly dependent: one of them can be removed (for instance, if one requires condition (i) from Definition IV.1, then suffices to constrain only sums of n-1 rows, whereas the remaining sum automatically equals k/n)). Whenever $k \le n$ and $n \ge 3$, one has $\frac{k(k+1)}{2} + n - 1 < kn$, which implies that the set nSt(n,k) has the dimension (k-1)n + 1 - k(k+1)/2.

Statements (i) and (ii) are immediate. Now regarding (iii), it is obvious that each row has norm $\sqrt{k/n}$ if and only if V can be written as (16). Notice now the columns are unit vectors if and only if $\sum_{m=1}^{n} \cos^2 \alpha_i = n/2 = \sum_{m=1}^{n} \sin^2 \alpha_i$, which in turn holds if and only if $\sum_m \cos 2\alpha_m = 2 \sum_m \cos^2 \alpha_m - n = 0$. Similarly, the columns are orthogonal if and only if $\sum_{m=1}^{n} \cos \alpha_i \sin \alpha_i = 0 = \frac{1}{2} \sum_m \sin 2\alpha_m$. These two constraints are equivalent to (17).

Proof of Lemma IV.2. The case where $\alpha = \beta = 0$ is trivial: Z = 0 and it obviously can be decomposed as in (18) with p = q = 0. Notice that every eigenvalue of $Z = Z^{\top}$ corresponds to the eigenvalue $\lambda^2 - 2\alpha\lambda$ of $Z^2 - 2\alpha Z$, and hence $\lambda^2 - 2\alpha\lambda - \beta = 0$. Therefore, $\alpha^2 + \beta \ge 0$ (otherwise, eigenvalues of Z would be complex). Furthermore, $\alpha^2 + \beta \ne 0$ (otherwise, $\lambda = \alpha$ would be the only eigenvalue of Z of multiplicity n, and one would have trace(Z) = αn , entailing that $\alpha = \beta = 0$). Denoting $\Delta = \sqrt{\alpha^2 + \beta}$, the matrix Z has two different eigenvalues $\alpha + \Delta$ and $\alpha - \Delta$, denote their multiplicities by k and n - k. Then $(\alpha + \Delta)k + (\alpha - \Delta)(n - k) = 0$. Denoting $q = \Delta - \alpha$ and $p = 2\Delta > 0$, one has (p - q)k - q(n - k) = 0or, equivalently, pk = qn thus, q > 0.

Consider the orthonormal eigenvectors v_1, \ldots, v_k , corresponding to the eigenvalue $p - q = \alpha + \Delta$ and orthonormal eigenvectors w_1, \ldots, w_{n-k} , corresponding to $-q = \alpha - \Delta$. The sequence $v_1, \ldots, v_k, w_1, \ldots, w_{n-k}$ constitutes an orthonormal basis of eigenvectors for the operator Z. Stacking the columns v_i and w_i , one obtains $n \times k$ and $n \times (n-k)$ matrices $V = (v_1, ..., v_k), W = (w_1, ..., w_{n-k})$. The matrix [V, W] is orthonormal and diagonalizes Z, that is, Z[V, W] = $[V,W] \mid (p-q)I_{FIXHERE}$ 0 and thus Z =[V, W] $\begin{bmatrix} 0 & -qI_{FIXHERE} \end{bmatrix}$ and thus $Z = (p-q)VV^{\top} - qWW^{\top}$. Since $VV^{\top} + WW^{\top} = I_n, Z$ is decomposed as (18). It remains to notice that $V^{\top}V = I_k$ by definition of the orthonormal basis and $\operatorname{diag}(VV^{\top}) =$ $(q/p)I_n = (k/n)I_n$ since, by (18), diag $(Z) = \mathbb{O}_{n \times n}$. To finish the proof, notice that $p-2q=2\alpha$ and $\beta=\Delta^2-\alpha^2=$ $(\Delta - \alpha)(\Delta + \alpha) = q(p - q).$

Proof of Corollary IV.3. Let $f(z) = z^2 - 2\alpha z, z \in \mathbb{C}$. It suffices to show that, if $f(Z) = \text{diag}(\beta_1 I_{n_1}, \dots, \beta_s I_{n_s})$, then $Z = \operatorname{diag}(Z_1, \ldots, Z_s)$, where $f(Z_i) = \beta_i I_{n_i}$. This statement will be proved for any analytic function f(z). It is well known that the spectrum of f(Z) consists of all points $f(\lambda)$, where λ is an eigenvalue of Z. Consider the set of eigenvalues of Z that belong to $f^{-1}(\beta_i)$ and let \mathcal{X}_i be the sum of corresponding eigenspaces. Then \mathcal{X}_i is invariant under the operator Z, and $\mathbb{R}^n = \bigoplus_{i=1}^s \mathcal{X}_i$ (the sum is orthogonal). Also, $f(Z)x = \beta_i x$ for any $x \in \mathcal{X}_i$. For any basis vector $e_r = (0, \ldots, 1, \ldots, 0)^+$ consider the decomposition $e_r = \bigoplus_{i=1}^s e_r^i, e_r^i \in \mathcal{X}_i$. Then $Ze_r = \bigoplus_{i=1}^s Ze_r^i, Ze_r^i \in \mathcal{X}_i \text{ and } f(Z)e_r = \bigoplus_{i=1}^s f(Z)e_r^i =$ $\bigoplus_{i=1}^{s} \beta_i e_r^i$. Suppose that $1 \leq r \leq n_1$. Then $f(Z)e_r = \beta_1 e_r$. Since β_1, \ldots, β_s are pairwise different, we have $e_r = e_r^1$ and $e_r^2 = \ldots = e_r^s = 0$. Similarly, for $n_1 + n_2 + \ldots + n_{j-1} + 1 \leq 1$ $r \leq n_1 + n_2 + \ldots + n_{j-1} + n_j$ one has $e_r = e_r^j$ $(j = 2, \ldots, s)$.

In other words, each \mathcal{X}_i contains n_i basis vectors e_r , where $n_1 + n_2 + \ldots + n_{i-1} + 1 \leq r \leq n_1 + n_2 + \ldots + n_{i-1} + n_i$ and thus dim $\mathcal{X}_i \geq n_i$. Recalling that $n_1 + \ldots + n_s = n$, one shows that dim $\mathcal{X}_i = n_i \forall i$ and thus \mathcal{X}_i is spanned by the corresponding basis vectors. Since \mathcal{X}_i is invariant under $Z, Z = \text{diag}(Z_1, \ldots, Z_s)$, where the block Z_i has dimension $n_i \times n_i$. Obviously, $f(Z_i) = \beta_i I_{n_i}$. The statement of Corollary is now immediate from Lemma IV.2.

Proof of Theorem IV.5. We prove the necessity first. Denoting $2\alpha = -\mathcal{D}(Z)$. By assumption, $Z^2 - 2\alpha Z$ is diagonal. Statements (i) and (ii) follow from Corollary IV.3, entailing also that p_i, q_i can be represented as (23) with some β_i . Since $Z_i^2 = 2\alpha Z_i + \beta_i I_{n_i}$ and diag $(Z_i) = \mathbb{O}_{n_i \times n_i}$, one has trace $Z_i^2 = \beta_i n_i$, therefore

$$\sum_{i=1}^{s} \beta_i n_i = \operatorname{trace}(Z^2) = 1.$$
⁽²⁹⁾

Recall also that for each *i* one has $p_i k_i = q_i n_i$ or, equivalently,

$$\frac{2k_i}{n_i} = \frac{\sqrt{\alpha^2 + \beta_i} - \alpha}{\sqrt{\alpha^2 + \beta_i}} = 1 - \frac{\alpha}{\sqrt{\alpha^2 + \beta_i}} \quad \forall i: p_i, q_i \neq 0.$$

(if $\alpha = 0$, one always has $p_i, q_i \neq 0$, otherwise it is possible that $\beta_i = 0$ and then $Z_i = 0$). This implies condition 3 ($\varepsilon =$ sign α) and allows to determine α, β_i . In the case where $\varepsilon \neq 0$ notice that $n_i - 2k_i \neq 0$ for any *i* such that $Z_i \neq 0$. Thus

$$\frac{\beta_i + \alpha^2}{\alpha^2} = \frac{n_i^2}{(n_i - 2k_i)^2} \Longleftrightarrow \beta_i = \alpha^2 \frac{4n_i k_i - 4k_i^2}{(n_i - 2k_i)^2}.$$

In view of (29), one obtains that

$$\alpha = \varepsilon \left(\sum_{i: Z_i \neq \mathbb{O}_{n_i \times n_i}} \frac{4k_i n_i (n_i - k_i)}{(n_i - 2k_i)^2} \right)^{-1/2}$$

which entails (24). In the case of $\alpha = 0$, one has $p_i = 2\sqrt{\beta_i}$, $q_i = \sqrt{\beta_i}$ for any *i*, and (29) implies that $\sum_i q_i^2 n_i = 1$. This finishes the proof of statement (iv).

The proof of sufficiency is similar. For any *i* such that $Z_i \neq 0$, the coefficients p_i, q_i have the form (23) (if $\varepsilon \neq 0$, this is implied by (iv)a, otherwise we choose $\alpha = 0$ and $\beta_i = q_i^2 = p_i^2/4$). Therefore, we have $Z_i^2 - 2\alpha Z_i = \beta_i Z_i$ and, in particular, $Z^2 - 2\alpha Z$ is diagonal. A straightforward computation shows that $p_i k_i = q_i n_i$ and thus diag $(Z_i) = \bigcup_{n_i \times n_i} \forall i$,

in particular, diag $(Z) = \mathbb{O}_{n \times n}$. Also, diag $(Z_i^2) = \beta_i I_{n_i}$, and statement (iv) now implies that trace $Z^2 = 1$. It remains to notice that $Z_i^3 = 2\alpha Z_i^2 + \beta_i Z_i$, and hence trace $(Z_i^3) = 2\alpha \beta_i n_i$. Hence, $\mathcal{D}(Z) = -\text{trace}(Z^3) = -2\alpha, Z^2 + \mathcal{D}(Z)Z$ is a diagonal matrix, and Z is an equilibrium (14).

APPENDIX B SCALE-SYMMETRIC MATRICES

We now generalize our results for symmetric appraisal networks to a class of asymmetric matrices. We define the sets of *scale-symmetric* matrices

 $\mathbb{R}^{n \times n}_{\text{zero-diag,dss}} = \{ A \in \mathbb{R}^{n \times n}_{\text{zero-diag}} \mid \text{there exists } \gamma \succ \mathbb{O}_n \text{ such that} \\ A \operatorname{diag}(\gamma) = (A \operatorname{diag}(\gamma))^\top \}, \\ \mathbb{S}^{n \times n}_{\text{zero-diag,dss}} = \mathbb{S}^{n \times n}_{\text{zero-diag}} \cap \mathbb{R}^{n \times n}_{\text{zero-diag,dss}}.$

Note that $\mathbb{S}_{\text{zero-diag,dss}}^{n \times n} \supset \mathbb{S}_{\text{zero-diag,symm}}^{n \times n}$ and

$$\begin{split} \mathbb{S}_{\text{zero-diag,dss}}^{n \times n} &= \bigcup_{\gamma \succ 0_n} \mathbb{S}_{\text{zero-diag,dss}}^{n \times n}(\gamma), \\ \mathbb{S}_{\text{zero-diag,dss}}^{n \times n}(\gamma) &= \{A \in \mathbb{S}_{\text{zero-diag}}^{n \times n} \mid A \operatorname{diag}(\gamma) = (A \operatorname{diag}(\gamma))^\top \end{split}$$

Lemma B.1. Consider any $\gamma \succ \mathbb{O}_n$ and some matrix $A \in \mathbb{R}^{n \times n}$ such that $A \operatorname{diag}(\gamma) = \operatorname{diag}(\gamma) A^{\top}$. Then,

- (i) A has real eigenvalues and it is diagonalizable,
- (ii) $\operatorname{trace}(A^2) = 0$ if and only if A = 0.

Proof. Since $A \operatorname{diag}(\gamma)$ is symmetric, then $A' = \operatorname{diag}(\gamma)^{-1/2} A \operatorname{diag}(\gamma)^{1/2}$ is also symmetric and thus has real eigenvalues and its eigenvectors form an orthogonal basis. Now, let (λ, v) be an eigenpair for A'. Then, by defining $u = \operatorname{diag}(\gamma)^{1/2} v$, we observe that $Au = \lambda u$, and so $(\lambda, \operatorname{diag}(\gamma)v)$ is an eigenpair for A. Hence the eigenvectors of A form a basis, and thus A is diagonizable. This proves (i). Observe that $A = \operatorname{diag}(\gamma)A^{\top}\operatorname{diag}(\gamma)^{-1}$. Then, $\operatorname{trace}(A^2) = \operatorname{trace}(A \operatorname{diag}(\gamma)A^{\top}\operatorname{diag}(\gamma)^{-1})$. From simple algebraic operations, it can be found that $\operatorname{trace}(A^2) = 0$ if and only if A = 0. This proves (ii).

In view of Lemma B.1, a matrix A is scale-symmetric if and only if $A = D^{-1}A_sD$, where D > 0 is a positive diagonal matrix (in Lemma B.1, $D = \text{diag}(\gamma^{-1/2})$ for some $\gamma \succ \mathbb{O}_n$) and A_s a symmetric matrix.

Recall the invariance property of the pure-influence model (2): if $X(0) = X(0)^{\top}$, then $X(t) = X(t)^{\top}$ for all t > 0. We are now ready to provide a more general version of this property: If D > 0 is a diagonal matrix and X(t) is a solution, then $DX(t)D^{-1}$ is also a solution. For this reason, if $X(0) = DX_s(0)D^{-1}$ is a scale-symmetric matrix with some $X_s(0) = X_s(0)^{\top}$, then the solution $X(t) = DX_s(t)D^{-1}$ is scale-symmetric. A similar result holds for the projected pure-influence model (9). Indeed, all of the theoretical results obtained in this paper for symmetric appraisal matrices. For example, if $X(0) \in \mathbb{R}_{\text{zero-diag,dss}}^{n \times n} (Z(0) \in \mathbb{S}_{\text{zero-diag,dss}}^{n \times n} (S_{\text{zero-diag,dss}}^{n \times n})$.



machine learning

Pedro Cisneros-Velarde Pedro Cisneros-Velarde received the B.Sc. in Electrical Engineering at the Pontifical Catholic University of Peru (PUCP), Lima, Peru; a M.Sc. in Control Theory and Network Science and a M.A. in Statistics at the University of California, Santa Barbara (UCSB), CA, USA. He is currently a PhD student at UCSB with the Center for Control, Dynamical Systems, and Computation and the Department of Electrical Engineering. His broad research interests include the areas of multiagent systems, game theory, networked systems and



}.

Noah E. Friedkin is with the University of California at Santa Barbara, where he is a professor of sociology and a faculty member of the Center for Control, Dynamical-Systems, and Computation. He moved to Santa Barbara after receiving the B.A. and Ph.D. from the University of Chicago. He is Editor-in-Chief of the Journal of Mathematical Sociology and a AAAS Fellow. Other honors include a James S. Coleman Distinguished Career Achievement Award and two Harrison White Outstanding Book Awards from the Mathematical Sociology Sec-

tion of the American Sociological Association. His publications concentrate on the investigation of social influence systems.



Anton Proskurnikov (M'13, SM'18) is an assistant professor at the Department of Electronics and Telecommunications, Politecnico di Torino (Italy) and a part-time senior researcher at the Institute for Problems in Mechanical Engineering of the Russian Academy of Sciences. He received the M.Sc. and Ph.D. degrees in applied mathematics from St. Petersburg State University in 2003 and 2005, respectively. Previously, he was an assistant professor at St. Petersburg State University (2003-2010), a postdoctoral researcher at the University

of Groningen (2014-2016) and Delft University of Technology (2016-2018). Anton Proskurnikov is an associate editor in IEEE Transactions on Automatic Control and Journal of Mathematical Sociology. His research interest include network science, multi-agent systems and technologies, nonlinear dynamics, control applications in social and biological sciences.



Francesco Bullo (S'95-M'99-SM'03-F'10) is a Professor with the Mechanical Engineering Department and the Center for Control, Dynamical Systems and Computation at the University of California, Santa Barbara. He was previously associated with the University of Padova, the California Institute of Technology, and the University of Illinois. His research interests focus on network systems and distributed control with application to robotic coordination, power grids and social networks. He is the coauthor of "Geometric Control of Mechanical

Systems" (Springer, 2004) and "Distributed Control of Robotic Networks" (Princeton, 2009); his "Lectures on Network Systems" (KDP, 2019) is available on his website. He is a Fellow of IEEE, IFAC, and SIAM.